 CHOHAN /22-August-05

Week 34 starting 22 Aug 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  22 August 05 : 8/22/2005 4:19:42 PM  
	7
	A1
	MBBL2213
	PT 5.1 1.5 KA Quench LF
	1.90

	1
	A2
	MBAL1084
	PT 12 Warm Up
	11.08

	3
	B1
	LE-C11R7
	Thermal cycle - No Tcoils
	145.75

	10
	B2
	MBAR3374
	ICS 4 Final connection
	295.62

	4
	C1
	MBAL3377
	PT 6.3 Training 3
	12.74

	9
	C2
	MBBR3384
	ICS 4 Final connection
	295.18

	11
	D1
	SSS245
	PT 13.2 Resist. Meas.
	297.59

	12
	D2
	SSS607
	PT 12 Warm Up
	294.1

	8
	E1
	MBAL2205
	Prep 5 Cool Down
	103.58

	2
	E2
	MBAL2072
	PT 12 Warm Up
	262.94

	6
	F1
	MBBR3387
	Prep 5 Cool Down
	103.58

	5
	F2
	MBBL2200
	PT 6.1 Training 1
	1.93 HV


  Tempr. from TBSM18 site
   
Coordinator this week: marco; MTM support 16-3602

CC on B1 since 3rd Aug; 1084 on A2 since 10th Aug
Generally very good week for Dipoles…particularly end of last week with fast turn-round & good perf.     ICS ran out of dipoles Sunday afternoon!
12 complete Magnet Tests cycles ( +1 not counted last week SSS607 TC)
Only 11 dipoles ( 0 with MM )+ 1 SSS(without MM & SSW )+1 ssss (607)  Thermal cycle previous week was not counted
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) Last week: Many Thermal cycles (Ansaldo) ; SSS607 (Q4) on D2 MM tests last week; CC on B1 long cooling down… 1084 on A2 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
·  Special  Points :

· Special SSS 606(Q5) on D2  (J Halik set up after Tuesday this week)
· 1084  Lifetime tests : 200A/s  4000 cycles  finished  & Thermal cycle
· Connect. Cryostat Tests B1  : not arrived at 1.9 K ? (JPLamboy, JP Tock ?) 

Tests 

	Magnets under Test from 14-08-2005 to 20-08-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBL3138 
	[image: image1.png]



	


	11-08-2005 
	27 h.
	23 h.
	15 h.
	29 h.
	46 h.
	17-08-2005 
	140 h.
	Stand By
	
	

	2
	TBB2
	MBAL2190 
	[image: image3.png]



	


	09-08-2005 
	53 h.
	36 h.
	18 h.
	22 h.
	38 h.
	16-08-2005 
	167 h.
	Stand By
	
	

	3
	TBC1
	MBAL2532 
	[image: image5.png]



	


	11-08-2005 
	25 h.
	23 h.
	16 h.
	17 h.
	19 h.
	15-08-2005 
	100 h.
	Stripping
	
	

	4
	TBC1
	MBAR3353 
	[image: image7.png]



	


	15-08-2005 
	37 h.
	23 h.
	20 h.
	15 h.
	6 h.
	19-08-2005 
	101 h.
	Stripping
	
	

	5
	TBC2
	MBAL2182 
	[image: image9.png]



	


	13-08-2005 
	1 h.
	18 h.
	15 h.
	16 h.
	35 h.
	17-08-2005 
	85 h.
	Stripping
	
	

	6
	TBD1
	SSS124 
	[image: image11.png]



	


	12-08-2005 
	37 h.
	14 h.
	24 h.
	10 h.
	27 h.
	17-08-2005 
	112 h.
	Stand By
	
	

	7
	TBE1
	MBAL3106 
	[image: image13.png]



	


	13-08-2005 
	52 h.
	62 h.
	35 h.
	10 h.
	16 h.
	20-08-2005 
	175 h.
	Stripping
	
	

	8
	TBE2
	MBAL2208 
	[image: image15.png]



	


	14-08-2005 
	37 h.
	20 h.
	26 h.
	14 h.
	12 h.
	19-08-2005 
	109 h.
	Stripping
	
	

	9
	TBE2
	MBBL1191 
	[image: image17.png]



	


	08-08-2005 
	41 h.
	39 h.
	26 h.
	17 h.
	32 h.
	14-08-2005 
	155 h.
	Stripping
	
	

	10
	TBF1
	MBBL3075 
	[image: image19.png]



	


	09-08-2005 
	41 h.
	33 h.
	19 h.
	12 h.
	15 h.
	14-08-2005 
	120 h.
	Stripping
	
	

	11
	TBF1
	MBBR2220 
	[image: image21.png]



	


	14-08-2005 
	52 h.
	39 h.
	20 h.
	15 h.
	15 h.
	20-08-2005 
	141 h.
	Stand By
	
	

	12
	TBF2
	MBAR3348 
	[image: image23.png]



	


	12-08-2005 
	48 h.
	21 h.
	76 h.
	16 h.
	8 h.
	19-08-2005 
	169 h.
	Stripping
	
	


Also Finished 1153 on C2 & 1237 on B2  on Sunday 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL3138
	12272 A.
(D1-L-S13)
	12716.7 A.
(D2_L_S01)
	[image: image25.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL2190
	[image: image26.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAL2532
	11348 A.
(D1-L)
	12541 A.
(D1-L)
	12739 A.
(D1-L)
	0 A.
(No Data)
	[image: image27.png]



	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAR3353
	12171 A.
(D2_L)
	12851 A.
[image: image28.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL2182
	11794 A.
(D1-L)
	12306 A.
(D1-L)
	[image: image29.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS124
	11716.5 A.
(QI_12)
	12448 A.
(QI_12)
	12603.7 A.
(QI_34)
	[image: image30.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAL3106
	9732 A.
(D1_U)
	11988 A.
(D1_U)
	12800 A.
(D1-U)
	12850 A.
[image: image31.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAL2208
	10982 A.
(D1-L)
	11827 A.
(D1_U)
	12392 A.
(D1_U)
	12850 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL1191
	12421 A.
(D1-U)
	12617.8 A.
(D2-L)
	12850 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL3075
	10850 A.
(D1-U)
	12721.6 A.
(D2_U)
	12850 A.
(D2-L)
	[image: image34.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBR2220
	11815 A.
(D2-U)
	12850 A.
[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAR3348
	11368 A.
(D1_U_S13)
	11714 A.
(D1_U_S01)
	12809 A.
(D1-L-S01)
	12850 A.
[image: image36.jpg]



	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-


[image: image38.png]Magnet Summary

175

Hours

A S





	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _16_M
	TBF2


	Unable to launch MMP (shaft position error) while trying to perform MM1 on MB3348.
	MTM
	
	Maryline informed. She finds the sliding ruler for the position check (MM or Qloc) is wrongly adjusted

	Tuesday _16_A
	TBF2


	Over-range in sector0203 ap2 during MM1.
	MTM
	
	Guy asked to replace TRU2 by TRU4.

	Tuesday _16_N
	TBF2


	Over-range in sector 0001 ap2 during MM2.
	MTM
	
	Stephane informed Guy.

TRU4 changed with TRU5.

	Wednesday _17_N
	All


	Cryo  OK lost in all cluster.
	ACR
	
	

	Wednesday _17_N
	TBA2


	Cryo  OK lost caused by  Tleads fault.
	ACR
	
	

	Wednesday _17_N
	TBC1


	Broken connector in the cable HVrack -PLCSE
	MTM
	
	HV test. delayed. Repaired by F.Flamand 

	Thursday_18_M
	TBF2


	Panel on the top of F2 bench is not showing light when current is on
	MTM
	
	

	Thursday_18_N
	TBD2


	SSS607: quench during load-line
	AB/PO
	
	Quemch occurred during a SPA because the threshold of the potaim card wasnot adapted for the Q4 (higher inductance). Threshold changed on the potaim card.

	Thursday_18_N
	TBE1


	MXI error : 3rd training on MB3106 could not be launched.
	AB/CO
	
	LF and HF crates were reset and tried launching again but same error
Local server was logged out and logged in 
Hf and lf could arm.

	Thursday_18_N
	TBE1


	3rd training on MB3106 performed: No hf file..
	AB/CO
	
	Quench analysed using lf DAQ.

	Thursday_18_N
	TBE1


	E traveller was signed by ICS for standby/stripping without generating report for WP07 HV test.
	ICS
	
	Process stopped waiting for HV test result.

	Friday_19_M
	TBE2


	HF arming error on PT 6.4 in the MB3106. Magnet reached ultimate, but could not generate the u file.
	AB/CO
	
	Guy asked to perform the SPA without u file. He contacted A.Raimondo. He checked the crate and solved the problem.

	Friday_19_M
	TBF1


	MB2220: PT9 failed for the step QH versus Ground for 3050 V.
	MTM
	
	Investigation sequence launched. Then OK.

	Friday_19_A
	TBF


	Broken HV connector  between PLC HV and PLCSE
Same problem Sunday night.
	MTM
	
	2 hours to find a good contact. Repaired by F.Flamand.



	Friday_19_A
	TBA2


	Quench during special cycles in the 1084 magnet.
	AB/CO
	
	Giorgio found the mechanical switch open, he closed it

	Friday_19_A
	TBC2


	Application stuck in the bench.
	AB/CO
	
	Stop all and restart applications..

	Friday_19_N
	TBD2


	Full load line on SSS607: Error due to power supply or cryo OK: test aborted after two positions.
	?
	
	Guy came. Measurement was started again by launching the test for the remaining 4 position after modifying the run type.

	Friday_19_N
	TBD2


	Full load line on SSS607: test stop because of an error due to power supply or cryo OK. ( 2 times) 
	?
	
	Run recovery launched.

	Friday_19_N
	TBC2


	PT9 HV test was failing in the dipole-Quench- heater vs ground step at 3050V several times for MB1153.
	?
	
	The test was successful after re-launching the test 2-3 times.

	Friday_19_N
	TBC2


	PT6.3 on MB1153: Hf acquisition file generated but not lf file
	AB/CO
	lf
	

	Saturday_20_M
	TBE2


	MB2072: leak in the manchette.
	CRI
	
	.

	Saturday_20_A
	TD1


	HV failed in the step Orbit Dipoles Vs GND.
	MTM
	
	Condensation on orbital dipole connectors. Heater fan is missing.

Action P.Legrand?

	Sunday _21_M
	TBF


	PLCSE cable contacts loose
	MTM
	
	

	Sunday _21_N
	TBB2


	IAP @ 300K: lf heater discharge not OK.
	MTM
	lf
	No magnet (MB3374) in the DB because HWREC was off for the special test on TBB1.

Guy called.

HWREC switched on. File could be generated.



	Monday _22_M
	TBA2


	HV test failed on dipole vs ground .Moisture was seen on top of CFB.
	ACR
	
	Action taken by ACR team

	Monday _22_M
	TBD1


	PT13 (resistance and voltage measurements) on SSS245: errors on

-voltage of six Vtaps

-voltage of correctors (EF circuit)

-voltage of correctors (CD circuit)
	MTM
	
	Values different and correct as per template. Discrepancies coming from

· connection of power supply (-ve) to the meter -ve terminal which should be disconnected for the differential measurement
· reverse polarity connection (EF and CD) 


Green Connect + Pump down /Blue :  Wait + Cool down	Ivory: Tests at Cold (36 hr)Red :  Wait + Warm up (12 hr)	  Dark Green: Disconnect (8 hr)











